Author's response to reviews

Title: Update: cohort mortality study of workers highly exposed to polychlorinated biphenyls (PCBs) during the manufacture of electrical capacitors, 1940-1998

Authors:
Mary M. Prince (io12@earthlink.net)
Misty J. Hein (zcr9@cdc.gov)
Avima M. Ruder (amr2@cdc.gov)
Martha A. Waters (maw0@cdc.gov)
Patricia A. Laber (pal0@cdc.gov)
Elizabeth A. Whelan (eaw0@cdc.gov)

Version: 3 Date: 22 May 2006

Author's response to reviews:

Style revisions--citations and references + affiliations for all authors (note: all authors either current or former CDC employees. None has other current research affiliation).